Ref: your mail Re: 1566326679108104. Mortality attributable to tobacco: Review of different methods

Sir,

Please find enclosed the revised manuscript now titled “Mortality attributable to tobacco: Review of different methods”

We highly appreciate these comments for their usefulness in helping us make the content of the article more precise and clearer. Following are the detailed responses to these comments.

Reviewer: Katrien Vanthomme

Reviewer's report:

Discretionary revisions

Since the authors have agreed upon all remarks of the reviewers, I agree with the publication of the article. One last spelling check however is advisable.

R; it was done